Princeton University COS425 - Database Systems William O’Connell - Spring 1996

Homework #1 - E-R Diagrams

Graded out of 100 points; Due: in class, Tues., Feb 13th.

1. [30 pts] Construct an E-R diagram for a car insurance company with a set of customers each of which owns a number of cars. Each car has a number of recorded accidents associated with it.

2. [30 pts] Turn the following E-R diagram into relational tables:

3. [40 pts] Solve exercise 3.22 on pp. 66 of the textbook.

Job Applicant

Available Position

Department

Interview

Opening name salary-req date interviewer rating pname salary open-date

dname floor

1 n m n